

G OPEN ACCESS

Citation: Burgers EJ, Sharma RP, Eugenio CJS, Heldring MM, Wijaya LS, van de Water B, et al. (2025) Computational modelling identifies primary mediators of crosstalk between DNA damage and oxidative stress responses. PLoS Comput Biol 21(3): e01012844. https://doi.org/10.1371/journal.pcbi.1012844

11(ps.//doi.org/10.1071/journal.pcbl.1012044

Editor: Melissa L. Kemp, Georgia Institute of Technology and Emory University, UNITED STATES OF AMERICA

Received: July 18, 2024

Accepted: February 3, 2025

Published: March 10, 2025

Peer Review History: PLOS recognizes the benefits of transparency in the peer review process; therefore, we enable the publication of all of the content of peer review and author responses alongside final, published articles. The editorial history of this article is available here: https://doi.org/10.1371/journal.pcbi. 1012844

Copyright: © 2025 Burgers et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data availability statement: All code to

RESEARCH ARTICLE

Computational modelling identifies primary mediators of crosstalk between DNA damage and oxidative stress responses

Elsje J. Burgers, Raju P. Sharma, Carl Joshua S. Eugenio, Muriel M. Heldring, Lukas S. Wijaya, Bob van de Water, Joost B. Beltman¹

Division of Cell Systems and Drug Safety, Leiden Academic Centre for Drug Research, Leiden University, Leiden, The Netherlands

* j.b.beltman@lacdr.leidenuniv.nl

Abstract

Cells exposed to toxicants, such as drugs, activate a wide variety of stress pathways, often simultaneously. Two important pathways that can influence cell fate and consequently adverse reactions are the oxidative stress response (OSR) and the DNA damage response (DDR). Previous studies have presented evidence of crosstalk between the OSR and DDR. We aimed to develop computational models to describe experimentally observed dynamics of both OSR and DDR proteins in liver (HepG2) cells in vitro upon exposure to various concentrations of either diethyl maleate (DEM; an agent primarily invoking oxidative stress) or etoposide (an agent primarily causing DNA damage). With these models, we aimed to identify the key interactions that cause crosstalk and their importance in describing protein dynamics. We developed a new model for the OSR pathway, coupled it to a previously developed model for the DDR pathway, and extended the resulting combined model based on multiple potential modes of crosstalk described in the literature. The different models were applied to previously published data of HepG2 GFP-reporter cells with time-dynamic information on the relative amount of proteins important for the OSR (NRF2, SRXN1) or DDR (p53, p21, BTG2 and MDM2). The developed models properly described key OSR and DDR protein dynamics, and in silico knockdowns of key model components in most cases led to a moderate effect on the connected pathway. The largest effect occurred after knockdown of p21, which resulted in a substantial decrease in NRF2 and SRXN1. We expect these models could play a role in adversity predictions by coupling our models with other models that predict cell fate or adversity based on the expression of specific proteins.

simulate the models and generate the figures is available at

https://doi.org/10.5281/zenodo.14844666.

Funding: This project has received funding from the ZonMW InnoSysTox program under grant agreement no. 114027005 (SysBioToP-Moving; to BvdW and JBB). The funder had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing interests: The authors have declared that no competing interests exist.

Author summary

When cells are exposed to toxic substances, they can activate various pathways to regain homeostasis. Two important pathways include the oxidative stress response and the DNA damage response, which are also known to interact. In this study, we developed mathematical models combining the two pathways and including relevant interactions. We exploited data on key proteins for each pathway, acquired in cells originating from liver. The cells were exposed to etoposide (a DNA damage-inducing compound) or diethyl maleate (an oxidative stress-inducing compound). By calibrating multiple model variants to the data, we identified the key interactions between the two pathways. In the future, our models could also contribute to improved toxicity predictions, because both pathways have a role in potential adversity of chemicals or drugs.

Introduction

Drug-Induced Liver Injury (DILI) is a frequent adverse drug reaction and a major problem for the pharmaceutical industry. Hepatic toxicity is the main reason both for why drugs fail during clinical trials [1] and for why drugs are withdrawn from the market [1–3]. Animal studies are widely used for toxicity testing, but the predictions based on these assays are moderately successful [4]. Moreover, there is a growing awareness of the ethical objections related to animal testing, and efforts are being made to reduce the number of test animals required for toxicity testing [5,6]. Therefore, there is a high demand for new or improved methods predicting DILI well and early in the drug development process. This includes amongst others imaging technologies, omics technologies, advanced *in vitro* models and computational modelling [7,8].

Artificial intelligence and computational modelling approaches will likely be essential for improving toxicity predictions, reducing the need for test animals as well as for laborious *in vitro* experiments [5]. Moreover, mathematical modelling can help to gain a mechanistic understanding of biological processes [9]. Systems of ordinary differential equations (ODEs) are especially suitable for modelling temporal changes in complex biological systems, such as the modulation of adaptive cellular stress responses, because they can translate mechanistic processes into a mathematical framework and are easy and inexpensive to numerically simulate [10].

When cells are exposed to chemicals, such as drugs, this can cause cellular stress. This stress activates specific stress pathways [11] that mainly intend to regain homeostasis, but when the stress is too severe, it can lead to cell death (e.g. apoptosis or necrosis) and consequently lead to DILI when this occurs in the liver [12]. In the past years, mathematical models of various cellular stress pathways have been developed [13,14]. Often multiple of such pathways simultaneously become active upon drug exposure. This can be due to drugs activating more than one pathway yet also due to crosstalk between pathways. To fully understand why drug exposure in some cases leads to adverse reactions and to predict adversity, we thus need to combine pathways, including their interactions.

Two important pathways that can influence cell fate and consequently play a role in DILI include the DNA damage response (DDR) [15,16] and the oxidative stress response (OSR) [17,18]. The DDR activates when DNA damage occurs either spontaneously or due to exposure to chemicals, radiation or ultraviolet light [19]. Depending on the type and severity of the damage a distinct response takes place, which is mainly regulated by p53. p53 is phosphorylated by DNA damage sensing kinases or their downstream targets [20,21]. Following

its phosphorylation, p53 acts as a transcription factor (TF) which leads to the transcription of many downstream targets such as p21, B-cell translocation gene 2 (BTG2) and mouse double minute 2 homolog (MDM2) [22]. MDM2 is a negative regulator of p53; complex formation of these proteins reduces the TF activity of p53 and promotes p53 degradation [23,24]. The interplay between MDM2 and p53 largely determines the dynamics of the DDR [24]. In the past decades, many different models have been developed to describe the DDR for varying applications and differing a lot in complexity [25,26]. Specifically, we previously developed an ODE model describing the cisplatin-induced DDR [27], based on HepG2 liver cells [27].

The second stress response pathway of interest, i.e., the OSR, reacts to oxidative stress which is caused by the accumulation of reactive metabolites (RM) or reactive oxygen species (ROS). OSR activation results in increased anti-oxidants, such as glutathione (GSH), to scavenge ROS or RM [28]. This protective pathway is activated by nuclear factor erythroid 2-like 2 (NRF2, gene name NFE2L2). NRF2 is constantly produced at a high rate and binds to Kelchlike ECH-associated protein 1 (KEAP1) in basal conditions [29–31]. Subsequently, NRF2 is ubiquitinated and degraded by the proteasome [31]. Upon oxidative stress, KEAP1 becomes modified and one of the two binding sites between KEAP1 and NRF2 dissociates as a latch, while the other remains bound as a hinge. [32–34]. Two-site binding is required for the rapid degradation of NRF2 [31,32]. Since oxidative stress causes one of the bounds to dissociate, this reduces the ubiquitination and thus NRF2 stabilises. Furthermore, NRF2 will translocate to and accumulate in the nucleus where it binds antioxidant response elements (AREs), which promote the transcription of various target genes such as sulfiredoxin 1 (SRXN1) and NAD(P)H quinone oxidoreductase 1 (NQO1) [35,36]. In the past years, several ODE models have been developed to describe chemically induced oxidative stress [37–39].

Previous studies have presented evidence of crosstalk between the OSR and DDR. Specifically, reported interactions include p21 inhibiting NRF2 degradation [40], phosphorylated p53 reducing the transcriptional activity of NRF2 [41], ROS causing DNA damage [42], NRF2-mediated transcription of MDM2 [43,44] and NQO1 stabilising p53 [45]. Computational modelling can help to indicate the key interactions among these and their importance in describing protein dynamics [46]. Moreover, when coupling stress pathway models to models predicting cellular adversity such as DILI, taking crosstalk into account could be essential for making proper predictions. With respect to OSR and DDR crosstalk, only one computational model has been created to study this interaction. Specifically, Pereira et al. [47] found that within the modelled p53-NRF2 signalling network the only crosstalk mechanism required to explain experimental measurements in breast-mammary epithelial cells included shared ROS induction. Because this model was only applied to stress induced by ROS, was developed for early time points only, and was developed for breast epithelial rather than liver cells, we need a different model for application to DILI predictions.

In the current study, we established mathematical crosstalk models between OSR and DDR. We developed a new model for the OSR pathway and employed a previously developed model for the DDR pathway by Heldring et al. [48]. The DDR model [48] was previously used to describe responses to DNA damaging agent cisplatin, and to extrapolate the responses between liver cell lines. We here applied the new OSR model and previously published DDR model to previously published liver HepG2 data of cellular responses to other compounds [49]. Next, we combined these two pathway models based on various potential modes of crosstalk found in the literature. Multiple variants of this p53-NRF2-pathway model differed in their abilities to describe chemically-induced stress pathway activation in HepG2 cells. Consequently, our models indicate the importance of the different modes of crosstalk and their effect on stress pathway activation.

Results

DDR model simulates activity in DNA damage reporters upon etoposide exposure

We started by investigating previously published etoposide data from Wijaya et al. [49], who exposed HepG2 liver cells with integrated BAC-GFP reporters to this compound and monitored responses of the labelled DDR related proteins (p53, MDM2, p21, BTG2) over time. Etoposide is a topoisomerase II inhibitor, leading to double-strand DNA breaks [50]. As expected, the data of the HepG2 GFP-reporter cell lines exposed to etoposide exhibited a clear induction of the DDR (Fig 1A). Based on the measured cell viabilities (S1 Fig.), we selected six out of eight concentrations (i.e. 0.5, 1, 2.5, 5, 10 and 25 μ M). We deemed the concentrations above 25 μ M (i.e. 50 and 100 μ M) as too toxic because the average cell count (S1A Fig) clearly decreased and the mean fraction of AnV-positive cells (S1B Fig) exceeded 0.15 during day 3 of the experiment. Note that the mean fraction of PI-positive cells remained below 0.05 (S1C Fig) and was not taken into account for the dose selection.

To simulate the DNA damage response, we used our previously published model [48] (Fig 1B; see Methods for details), which we originally developed to describe the DDR upon cisplatin exposure. The model parameters (see S1 Table.) were calibrated using data for the six selected concentrations of etoposide. The model described the data for p53, MDM2, BTG2 and p21 upon exposure to etoposide reasonably well (Figs 1A and S2), suggesting the general applicability of the model for DNA damaging compounds. The dose-response relationship is non-linear as can be seen from the model-estimated values for the initial stress levels for each concentration of etoposide (S3A Fig), which saturate with increasing etoposide dose.

When investigating the data of other available reporters (Fig 1C), we observed a dosedependent upregulation in NRF2 target SRXN1 as well. A potential explanation is that there is crosstalk between DDR activation and OSR elements. In conclusion, the model described the etoposide-induced DDR well. The model does not contain any OSR components, even though we observe a clear induction of SRXN1 upon etoposide exposure. Therefore, adding OSR induction to this model could be a valuable extension.

OSR model describes the early peak in NRF2 after exposure to DEM

Next, we investigated data of a typical oxidative stress inducer, diethyl maleate (DEM) [51, 52]. Based on the cell viability data (S4 Fig.) we decided to include all eight doses available in the HepG2 data of Wijaya et al. [49] (i.e. 2, 10, 20, 40, 80, 120, 160, 200 μ M). Indeed, even for the highest dose, the mean relative cell count did not decrease, the mean fraction of AnV-positive cells never exceeded 0.1, and the mean fraction of PI-positive cells remained below 0.03.

We then developed a simplified model for the OSR (Fig 2A; see Methods for details) in order to properly simulate the NRF2 and SRXN1 data upon DEM exposure. We considered two model variants, one including NRF2-mediated stimulation of KEAP1 [53] and one without. Following model parameter calibration based on the 8 DEM concentrations (S2 Table.), the simulations of the model with NRF2-mediated KEAP1 regulation exhibited a similar dynamics as the data (Figs 2B and S5, red line). Note that the stimulating link from NRF2 to KEAP1 was required to describe the quick rise and decrease observed in NRF2 upon DEM exposure: A model without that link performed clearly worse, especially for high concentrations (Figs 2B and S5, blue line; model with NRF2-dependent KEAP1 production - AIC: -1836, BIC: -1714; model without NRF2-dependent KEAP1 production - AIC: -818). As was the case for the DDR model following etoposide exposure, for the selected OSR



Fig 1. DDR model describes data from HepG2 cells exposed to etoposide. A) Simulations (red) of the DDR model for BTG2, MDM2, p21 and p53 combined with data (black line represents the mean, grey points the measurements per replicate) for these proteins after exposure to 2.5 and 25 µM etoposide. B) Schematic overview of the model for the DDR. C) Data for NRF2 and SRXN1 HepG2 reporter cell lines after exposure to six concentrations of etoposide. The colour represents the dose and the shaded area indicates the standard deviation across replicates.

https://doi.org/10.1371/journal.pcbi.1012844.g001



Fig 2. OSR model describes data from HepG2 cells exposed to DEM when NRF2 stimulates KEAP1 production. A) Schematic overview of the model for the OSR. B) Experimental data along with simulations of the OSR model with (red) and without (blue) NRF2-dependent KEAP1 production. The black line represents the mean, and the grey points the measurements per replicate after exposure to 20 and 200 μ M DEM. C) Data for MDM2 and p53 after exposure to eight concentrations of DEM. The colour indicates the dose and the shaded area indicates the standard deviation.

https://doi.org/10.1371/journal.pcbi.1012844.g002

model (i.e., with NRF2-mediated KEAP1 regulation) following DEM exposure the estimated values of the initial stress levels exhibited a non-linear dose-response relationship (S3B Fig).

The data of other HepG2 reporters exposed to DEM also exhibited interesting dynamics in both p53 and MDM2 (Fig 2C). Expression of p53 rapidly declined during the first 5–7 hours after exposure, followed by a gradual increase in expression. To check if the profound initial decrease was not caused by outlier replicates and/or concentrations, we investigated the the early time points of individual p53 replicates. However, all of the replicates show similar dynamics (S6 Fig.). Rather than exhibiting a decrease like p53, MDM2 initially increased for approximately 40 hours, succeeded by a gradual decrease. Downstream targets p21 and BTG2 also mildly responded to DEM, with an increase over time for both proteins and a later decrease for p21 (S7 Fig.).

Even though it is known that oxidative stress (due to e.g. reactive oxygen species (ROS)) can lead to DNA damage [42], this process cannot solely explain the experimentally observed dynamics. As shown in Fig 1, DNA damage generally leads to an initial increase in p53 followed by a decrease since MDM2 stimulates p53 degradation. Therefore, the initial decrease in p53 cannot be explained only by oxidative stress-induced DNA damage, suggesting other crosstalk plays a role as well. In conclusion, our model describes the OSR dynamics upon DEM exposure well, yet needs to be expanded with crosstalk to describe the behaviour of p53 and MDM2 as well.

Activity of p21 is sufficient to explain OSR activation upon etoposide exposure

Our observation of an increase in SRXN1 expression upon exposure to etoposide in Fig 1C suggested crosstalk between DDR and OSR elements within HepG2 cells. Indeed, multiple interactions reported in literature could cause this increase in OSR activity, as proxied by SRXN1 in our case. Specifically, the presence of p21 can lead to an increase in NRF2 activity and its downstream targets [40]. This is caused by p21 competing with KEAP1 for binding to NRF2: When p21 binds to one of the two NRF2 binding sites, NRF2 stabilises because it cannot be ubiquitinated [40]. As a consequence, proteins downstream of NRF2, such as SRXN1 become highly expressed.

Another mode of DDR-OSR crosstalk that could be important is the reduction of transcriptional activity mediated by NRF2 via phosphorylated p53 [41]. Specifically, activated p53 can bind to promoter elements inhibiting transcription of NRF2 target genes. Since we observed upregulation of SRXN1, an inhibition of transcription of SRXN1 caused by p53 is unlikely. However, inhibition of KEAP1 transcription by phosphorylated p53 could potentially play a role because low expression of KEAP1 leads to little inhibition of NRF2, which in turn results in a high SRXN1 expression.

We considered both modes of crosstalk separately, as well as jointly. Moreover, we considered etoposide itself not to induce oxidative stress. Model M-E1 contains the p21-mediated NRF2/SRXN1 expression (termed p21-NRF2), model M-E2 the p53-mediated KEAP1 inhibition (termed p53-KEAP1) and model M-E3 includes both modes of crosstalk (see coloured arrows in Fig 3A; equations in Methods). This resulted in a total of three variants of the DDR-OSR crosstalk model for etoposide. For each model variant, we only calibrated the extra crosstalk parameter(s) required for crosstalk (see new and modified values in S3 Table., the other values remain as in S1 Table. and column 4 of S2 Table.). Note that this also led to a change in the degradation parameter(s) of the modified equations, to ensure the model starts in steady state (see Methods for details).



Fig 3. Crosstalk models to describe etoposide-induced OSR activity. A) Schematic overview of the model for the DDR and OSR upon DNA damage induction. Pathways are connected by crosstalk via p21 and p53p. B) Simulations of four different model versions (coloured lines) for NRF2 and SRXN1 combined with data (black line represents the mean, grey points the measurements per replicate) for these proteins after exposure to 2.5 and $25 \,\mu\text{M}$ etoposide.

https://doi.org/10.1371/journal.pcbi.1012844.g003

Following parameter calibration to NRF2 and SRXN1 data upon etoposide exposure, all 3 DDR-OSR crosstalk model variants described the data reasonably well (Figs 3B and S8). Note that the decrease of the mean NRF2 data after the first timepoint is caused by the absence of data in the first 3 hours in replicate 1 that exhibits overall a slightly lower expression. Especially for low concentrations, simulations were highly similar across the 3 model variants. However, for high concentrations, differences were more apparent. Specifically, models M-E1 and M-E3 still resulted in similar fits, with model M-E3 remaining high at late time points which matched the experimental data well, while model M-E1 described the data at early time points well. Model M-E2 exhibited slightly different dynamics for SRXN1 than in the experimental data, when comparing the initial delay in SRXN1 increase, the subsequent steepness of the SRXN1 increase, and the SRXN1 behaviour during the last 10 hours. Based on these qualitative observations, selection of either model M-E1 or M-E3 was thus appropriate.

Apart from visually evaluating these models, we also compared the models quantitatively by computing the AIC and BIC. Model M-E1 had both the lowest AIC and BIC (Table 1) and could therefore be regarded as the best model. In conclusion, based on both qualitative observations (Figs 3 and S8) and statistical evaluation with AIC and BIC (Table 1), each of the three DDR-OSR crosstalk models is an improvement from the model without crosstalk. Model M-E1, including only p21-NRF2 crosstalk, was the most appropriate to describe this data based on information criteria. This analysis thus suggests that p21 forms the most prominent link by which DDR activity promotes OSR activity.

NRF2-mediated MDM2 production combined with ROS-dependent DNA damage describes DDR activation upon DEM exposure

The observation that HepG2 cells exposed to DEM exhibited unexpected early responses in p53 and MDM2 dynamics (Fig 2C) suggested that besides ROS-induced DNA damage additional OSR-DDR crosstalk affected p53 and MDM2 dynamics. Therefore, we searched for other potentially relevant effects of oxidative stress proteins on the DNA damage pathway. One such interaction might be the NRF2-mediated transcription of MDM2, which has been reported in murine embryonic fibroblasts and pancreatic cells [43,44]. The quick increase in NRF2 upon DEM exposure could have led to a quick increase in MDM2 and consequently to increased degradation of p53.

Another interaction described in literature is the stabilization of p53 by NQO1 [45,54,55]. Like SRXN1, NQO1 is a downstream target of NRF2 and it binds and stabilizes p53 but does not block MDM2-dependent p53 degradation [56]. This interaction could help to explain the increase in p53 after approximately 20 hours. Since NQO1 is lacking in our OSR model we utilized SRXN1 as a surrogate for NQO1. A gene coregulation network-based analysis of transcriptomics data in HepG2 clustered SRXN1 and NQO1 in the same network module [57].

	-	1 /	0 1	-			
Model	Crosstalk	Equations	Fitted parameters	Cost (RSS)	AIC	BIC	
M-E0	No crosstalk	2-8, 10, 12, 13	0	133.07	-2317.16	-2317.16	
M-E1	p21-NRF2	2-8, 10, 13, 14	1	45.19	-3500.98	-3495.97	
M-E2	p53p-KEAP1	2-8, 12, 13, 15	1	48.18	-3430.73	-3425.73	
M-E3	p21-NRF2 &	2-8, 13-15	2	45.62	-3488.56	-3478.55	
	p53p-KEAP1						

The equations defining each model can be found in the Methods. The cost, AIC and BIC were determined based on the data for NRF2 and SRXN1 (1098 data points).

https://doi.org/10.1371/journal.pcbi.1012844.t001

This indicates that SRXN1 and NQO1 are in general coregulated, and likely exhibit similar dynamics.

We included several combinations of the three mentioned interactions (termed ROS-DDR for ROS-induced DNA damage, NRF2-MDM2 for NRF2-mediated MDM2 production, and SRXN1-p53 for SRXN1 (NQO1)-mediated p53 stabilization) in models M-D1, M-D2 and M-D3 (S4 Table.; for equations see Methods). All model variants visually improved the description of the experimental data compared to absence of OSR-DDR crosstalk (Figs 4B and S9A), which was supported by AIC and BIC statistics (Table 2). Nevertheless, the result was not yet satisfactory, primarily because none of the simulations led to an early p53 decline (Fig 4B).

In order to understand why a strong initial p53 decline was not observed for our model variants, we studied the impact of MDM2-related parameters on p53. This analysis showed that for the current parameter values, only an extremely high level of MDM2 would lead to the desired p53 decline. For instance, when we simulated model M-D3 with an 8-fold increase in the maximal speed of NRF2-dependent MDM2 production, the correspondence to p53 data improved but this came at the expense of the MDM2 description (S10 Fig.). Since the increase in MDM2 is only modest in the experimental data, we hypothesised that the MDM2-dependent p53 interaction might be different in the setting with DEM-induced DDR compared to the etoposide-induced DDR. Therefore, we replaced the p53-MDM2 interaction term in the models with a Hill equation such that MDM2-induced p53 degradation achieves a maximum rate at high MDM2 concentrations, leading to models M-D4, M-D5, and M-D6. After recalibrating the corresponding parameters simultaneously with calibrating different combinations of crosstalk (see S5 Table.), models M-D4 and M-D6 managed to describe the initial decrease and late increase, while model M-D5 did not (Figs 4C and S9B; lines for M-D4 and M-D6 almost overlap). Consistently, the cost values and associated AIC and BIC values

Model	Crosstalk	Equations	Fitted parameters	Cost (RSS)	AIC	BIC
M-D0	No crosstalk	3-13	0	100.46	-6014.24	-6014.24
M-D1	NRF2-MDM2 &	4, 5, 7–13, 16, 17	5	77.29	-6530.73	-6502.71
	ROS-DDR					
M-D2	NRF2-MDM2 &	3, 5, 7–13, 17, 18	4	92.56	-6170.70	-6148.28
	SRXN1-p53					
M-D3	NRF2-MDM2,	5, 7–13, 16–18	6	78.92	-6486.93	6453.30
	ROS-DDR &					
	SRXN1-p53					
M-D4	Adapted MDM2-	7–13, 16, 17, 19,	11	63.45	-6914.82	-6853.17
	p53 degradation,	20				
	NRF2-MDM2 &					
	ROS-DDR					
M-D5	Adapted MDM2-	3, 7–13, 17, 20,	10	88.65	-6245.30	-6189.25
	p53 degradation,	21				
	NRF2-MDM2 &					
	SRXN1-p53					
M-D6	Adapted	7–13, 16, 17, 20,	12	63.47	-6912.31	-6845.05
	MDM2-p53	21				
	degradation,					
	NRF2-MDM2,					
	ROS-DDR &					
	SRXN1-p53					

Table 2. Comparison of the quality of models describing DEM exposure.

The equations defining each model can be found in the Methods. The cost, AIC and BIC were determined based on data for p53 and MDM2 (2008 data points).

https://doi.org/10.1371/journal.pcbi.1012844.t002



Fig 4. Crosstalk models to describe DEM-induced DDR activity. A) Schematic overview of the model for the DDR and OSR upon oxidative stress. Pathways are connected by crosstalk via oxidative stress, NRF2 and SRXN1. B) & C) Different model versions without (B) and with (C) adapted MDM2-dependent p53 degradation. Simulations (coloured lines) for MDM2 and p53 combined with data (black line represents the mean, grey points the measurements per replicate) for these proteins after exposure to 20 and 200 μ M DEM.

https://doi.org/10.1371/journal.pcbi.1012844.g004

were lowest for these model variants (Table 2). Since Model M-D4 has fewer parameters and lower AIC/BIC than model M-D6, we took this as the preferred model. In conclusion, following inclusion of Hill-type p53-MDM2 interactions, the presence of NRF2-mediated MDM2 production and ROS-mediated DNA damage are sufficient to explain the observed DDR.

In silico knockdowns predict strong effect of p21 on OSR activity

Following selection of models M-E1 and M-D4 as best models for etoposide and DEM exposure respectively, we noted that the selected models do not properly represent each of the individual replicates due to inter-replicate variability. We therefore investigated whether slight variation of model parameters could explain this variability by performing a sensitivity analysis (see Methods). Indeed, the effect of some parameters on the simulations of M-E1 (S11A Fig) and M-D4 (S12A Fig) was large. Varying the parameters with the most profound effect on the simulations by 20% helped to describe most of the variability between replicates (S11B and S12B Figs). Describing the variability was most successful for the directly induced proteins (i.e., p53 and MDM2 for etoposide, and NRF2 and SRXN1 for DEM).

Following the development of two separate models describing either DEM exposure causing DDR activity or etoposide exposure causing OSR activity, we next wanted to combine the different crosstalk mechanisms into one model. In principle, the slight OSR-mediated activation of the DDR upon DEM exposure might also feedback onto the OSR, and vice versa for etoposide exposure. Nevertheless, this secondary feedback might be too low to be biologically relevant in practice.

Ideally, we would have combined both models into one single model. However, the presence of compound-specific model elements such as the adapted term for MDM2-dependent p53 degradation precluded this. Therefore, we developed model M-CD for the combined crosstalk model upon DEM exposure and M-CE for the combined model upon etoposide exposure (without refitting or changing parameters). We hypothesised that the results would be very similar to models M-D4 and M-E1 respectively because the indirect activation of a pathway would be too minor to in turn affect the dynamics of the initially activated pathway.

Figs 5A, 5B and S13 show the results for models M-CE and M-CD. As expected, both for the comparison of models M-CE and M-E1 upon etoposide exposure (Figs 5A and S13A) and of models M-CD and M-D4 upon DEM exposure (Figs 5B and S13B) the results are indeed similar. The largest deviation occurs in the simulation of SRXN1 upon DEM exposure, yet the description of the data is still reasonable.

In order to further quantify the importance of the crosstalk mechanisms between OSR and DDR, we used the models M-CE and M-CD to perform *in silico* knockdowns. Specifically, we determined the expected effect of knocking down NRF2, p53, MDM2 and p21 in combination with exposure to either etoposide or DEM. To simulate knockdowns we multiplied the modelled production terms (both basal production and production dependent on other variables) by 0.3, i.e. a 70 % reduction (see Methods for details).

The knockdown predictions for the two models show similar trends (Figs 5C, 5D, S14 and S15): An MDM2 knockdown leads to a large increase in p53 and also to an increase in NRF2 and SRXN1, caused by the increase of p21 downstream of p53. In the results for model M-CD the increases are more profound than in those for model M-CE. A knockdown of NRF2 leads to a strong decrease in SRXN1, a smaller but clear decrease in MDM2 and consequently a slight increase in p53 in both models. A knockdown in p21 hardly affects MDM2 and p53 but does lead to a substantial decrease in NRF2 and SRXN1 (especially in M-CE). Lastly, the p53 knockdown has a similar but less strong effect (compared to p21 knockdown) on NRF2 and SRXN1. As expected, a p53 knockdown also leads to a decrease in MDM2. In conclusion, our models predict that knockdowns within either OSR or DDR have a limited to moderate effect on protein expression in the connected pathway via crosstalk. Especially p21 is expected to substantially affect OSR activity.



Fig 5. *In silico* **knockdown predictions on the basis of crosstalk models.** A-B) Simulation of best crosstalk model versions (blue or pink) and combined crosstalk models (red) for exposure to 25 µM etoposide (A) or 200 µM DEM (B). Simulations (coloured lines) and data (black line represents the mean, grey points the measurements per replicate) are shown for MDM2, NRF2, p53 and SRXN1. C-D) Simulations with (coloured lines) or without knockdowns (black line) for MDM2, NRF2, p53 and SRXN1, for exposure to 25 µM etoposide (C) or 200 µM DEM (D).

https://doi.org/10.1371/journal.pcbi.1012844.g005

Discussion

Cells exposed to toxicants, such as drugs, activate a wide variety of stress pathways, often simultaneously. While computational models for individual pathways exist, we aimed to

develop models to describe the dynamics of both OSR and DDR proteins simultaneously in HepG2 cells upon exposure to either etoposide or DEM. These models gave insight into the crosstalk between the two stress pathways. Our approach involved developing a simple model for the OSR pathway and adapting a previously developed model for the DDR pathway [48], and applying them to previously published data of Wijaya et al. [49]. Importantly, our models mainly contain key variables for which we had experimental data available, which implies that the models are simplified and do not include some of the relevant DDR and OSR processes. However, this approach allowed for parameter identification and provided us with comprehensible and computationally cheap models. Moreover, these base models successfully described the dynamics of key pathway proteins upon exposure to either DEM or etoposide. In the future, our models can be supplemented with additional processes taking place during the DDR and OSR, rendering a realistic description of the true complexity of these pathways.

Next, we coupled the two base pathway models, considering various potential modes of crosstalk found in the literature, considering the two crosstalk directions individually. By selecting the most suitable models to describe the data based on information criteria we managed to pinpoint which crosstalk mechanisms were relevant for HepG2 cells. For both etoposide and DEM exposure, this resulted in models decently describing the primary pathway activation as well as the secondary responses caused by crosstalk.

The sensitivity analysis that followed showed that varying certain parameters by 20% can largely explain the variability between replicates. This approach was most successful in describing the variability in the directly induced proteins (i.e., p53 and MDM2 upon etoposide exposure, and NRF2 and SRXN1 upon DEM exposure). For indirectly induced proteins, this approach was less successful: the absence of direct stress likely causes the parameters to have only a small effect. Rather, experimental noise is relatively large for these proteins as they have a lower intensity than upon direct induction.

To further explore the impact of the crosstalk, we included crosstalk in both directions in each of the models and performed *in silico* knockdowns. Both models led to similar trends upon knockdown, i.e., the same proteins increased or decreased compared to the situation without knockdown, providing confidence in our model-based expectations. However, the relative change for each protein differed between the simulation for DEM or etoposide exposure. The most profound effect of the crosstalk occurred during simulations of p21 knockdown, which led to a clear decrease in SRXN1 and NRF2.

To investigate the congruence between our models' knockdown predictions and experimental data, we compared our predictions to results of Hiemstra et al. [58]. Upon knockdown of p53, Hiemstra et al. [58] observed only a slight difference in SRXN1 after exposure to 10 or 100 µM DEM. Similarly, in our simulations a knockdown of p53 hardly affected SRXN1 for either concentration (S14B Fig). Furthermore, when Hiemstra et al. [58] measured p21 expression after knockdowns combined with exposure to etoposide, NRF2 knockdown hardly changed p21 but as expected p53 knockdown did decrease the p21 intensity significantly. This is in accordance with our simulations of p21 after etoposide exposure and NRF2 or p53 knockdown (S15 Fig.): The p53 knockdown had a strong decreasing effect, while the NRF2 knockdown had a slight increasing effect. Still, minor quantitative differences between earlier published data and our model predictions occur, partly because our models are simplifications, but also because our rough estimate of the knockdown efficacy (~ 70 % reduction of production rates) is uncertain. Nevertheless, the overall similarities between the in silico and in vitro knockdown results, indicate that we captured the main biological connections with our models. Future knockdown data, acquired in various reporter cell lines would be of value to test all our in silico predictions and to evaluate model validity.

During the model selection process, we selected models that excluded some crosstalk interactions found in the literature, because these mechanisms did not sufficiently improve the description of the HepG2 data. Specifically, we omitted the p53p-mediated inhibition of NRF2 transcriptional activity from the etoposide model, and the inhibition of p53 degradation via NQO1, from the DEM model. This might suggest that these modes of crosstalk are not that relevant for describing the HepG2 protein dynamics. However, we want to emphasise that the conclusions about the existence and relevance of crosstalk interactions heavily depend on the dataset that we used, which had limitations. For instance, for the first omitted interaction, i.e., the reduction in transcriptional activity mediated by NRF2 caused by p53p, the AIC/BIC analysis indeed led to our model selection decision not to include the interaction in our final model for etoposide. However, the dynamics of model M-E3 qualitatively matched the data slightly better than model M-E1 in terms of absence of SRXN1 decrease at late time points (Figs 3 and S8). If data at time points well beyond 60 hours had been available, the AIC/BIC analysis might have led to a quantitative preference for model M-E3 over M-E1.

The redundancy of the p53p-mediated crosstalk could be due to only limited transcriptional activity mediated by NRF2 in our data, compared to the results of Faraonio et al. [41]. Even though HepG2 cells upregulated NRF2 only slightly, a clear response in SRXN1 occurred. If the transcription of ARE genes were reduced, we would expect the transcription of SRXN1 to be affected as well. Faraonio [41] found the largest decrease in transcriptional activity when cells were exposed to both DEM and cisplatin, directly leading to oxidative stress and DNA damage, respectively. This could explain why the crosstalk effect is less profound when there is only DNA damage stress. Consequently, it might be required to reintroduce p53p-mediated inhibition of NRF2 activity in the models when applying them to a situation with both types of stress. Furthermore, Chen et al. [59] reported that p53 can have both a stimulating and an inhibiting effect on NRF2, depending on the amount of p53. Thus, the p53p-NRF2 interaction may be more complicated than only stimulation, which would require a different implementation in our models to correctly describe diverse biological settings.

Besides p53p-mediated NRF2 activity, other crosstalk mechanisms could play a role in the activation of the OSR as well. One of these mechanisms could be the activation of NRF2 via BTG2. Specifically, BTG2 mediates upregulation of antioxidant enzymes known to be regulated by NRF2 in MCF7 cells [60]. BTG2 may act as a transcriptional co-activator for NFE2L2 but the effect could also be NRF2-independent. Even though model M-E1 is already able to describe the data well, these prior findings warrant further investigation of these interactions in HepG2 cells. Furthermore, etoposide was recently reported to induce oxidative stress in bacteria (Pseudomonas aeruginosa) [61] and DNA damage has been reported to induce ROS in both yeast and mammalian cells [62,63]. Even though it is unclear if this also happens in HepG2 cells, it challenges our assumption that etoposide exposure does not lead to oxidative stress. Furthermore, other signalling pathways could be involved in the crosstalk between DDR and OSR. For example, NF- κ b is known to have both pro- and anti-oxidative targets [64,65] and to be activated in the DNA damage response [66,67]. Similarly, JNK activation and redox signalling is linked to DNA damage [67,68] and consequently affects OSR activation [69,70]. Additional experiments should further elucidate the mentioned processes and could confirm if it is required to expand our model with etoposide-driven ROS production or additional signalling pathways.

With respect to crosstalk from OSR to DDR, model fits did not improve upon inclusion of p53 stabilization via NQO1. Here, one potential caveat is the replacement of NQO1 by SRXN1 in our model implementation, which was required due to the lack of NQO1 data.

Since NQO1 and SRXN1 may exhibit differences in their dynamics, it is possible that including the NQO1 crosstalk in the model via SRXN1 prohibited improvement of fit quality. It should also be noted that the basal level of NQO1 in HepG2 and other hepatocellular carcinoma cells is relatively high [71,72]. Consequently, the effect of NQO1 might not be captured correctly by taking SRXN1 as an NQO1 proxy. The impact of NQO1 on p53 could thus be more important in other, non-cancerous, cell lines. Additional data on the protein dynamics of NQO1 would help to evaluate the importance amongst OSR-DDR crosstalk.

Based on information criteria analysis, we selected model M-D4 as appropriate model to describe the induction of p53 and MDM2 upon DEM exposure (Fig 4 and Table 2). However, the description of these DDR data could be further improved: Unlike the data, the simulations of MDM2 for high DEM concentrations, are increasing rather than decreasing after approximately 30 hours. Initially, we expected the decrease of MDM2 would be due to the rapid decrease in NRF2, since NRF2 is one of the stimulators of MDM2 in this model. However, this did not happen in the model, either because the increase in p53p compensated for the decrease in NRF2 or because the effect of NRF2 on MDM2 is only small. Refitting the parameters for the MDM2 production might lead to a closer match, but this would also worsen the simulations for exposure to etoposide.

Besides a crosstalk effect of NRF2 on MDM2, alternative mechanisms may explain high MDM2 degradation at late time points, as observed experimentally. DNA damage can lead to increased auto-degradation of MDM2 [73]. Since oxidative stress can induce DNA damage [42], as is included in our model, MDM2 autodegradation could be a reason for the decrease in MDM2 levels at late time points. One might expect this effect to also apply to the data for etoposide, where MDM2 only increased. However, the levels of p53 are very high following exposure to etoposide, and the associated high production rate of MDM2 might compensate for an increased auto-degradation. Therefore, we cannot exclude that MDM2 auto-degradation plays a role during the DEM-induced DDR.

A complicating factor in improving the p53/MDM2 part of the model is the negative feedback loop between MDM2 and p53. Changing MDM2 affects p53 and vice versa, which makes it difficult to adapt the model for both proteins simultaneously. Indeed, the relation between MDM2 and p53 is probably more complex than we currently included in the model, as evident from the difficulty to properly simulate the initial decrease in p53 upon exposure to DEM. We achieved this by adding compound-specific MDM2-dependent p53 degradation, which suggests that additional processes play a role in this difference in interaction between p53 and MDM2 across compounds. One complicating factor in quantifying the effect of MDM2 on p53 is the existence of MDMX (also known as MDM4), a homolog of MDM2 [74]. MDMX modulates the MDM2-dependent degradation of p53, and MDM2 and MDMX mutually depend on each other for the regulation of p53 [75]. Moreover, the regulation of both MDM2 and MDMX is complex: Even though both are transcriptionally activated by p53, they are also differentially regulated in many ways [76]. Thus, additional data on MDMX are required to understand if and how much effect it has in the settings studied here.

In general, model simplification by adapting the value of parameters depending on the compound, is a straightforward and potentially useful solution to obtain proper simulations of biological data. In this case, changing a few parameter values helped to describe the initial decrease in p53 correctly upon DEM exposure. However, such an approach also comes with disadvantages: Due to the compound-specific parameters in our models, we could not combine the two separate models into a single joint model. Moreover, having compound-specific parameters implies that such parameters need to be recalibrated when applied to data for new compounds unless the compound has highly similar effects as a compound previously used for parameter calibration. This reduces the applicability and generalisability of

models. Therefore, further research into the mechanisms leading to compound specificity in parameters is highly desirable.

Our models are applicable in a certain domain: for specific concentrations of DEM and etoposide, they can describe the response for up to 60 hours. In future research, it would be relevant to investigate how well these models perform outside these conditions. By generating a dataset for different concentrations or longer timescales, one could investigate the extrapolation capacity of these models. It should be noted that the 2D cell culture conditions for these experiments preclude imaging far beyond 60 hours. Hence, 3D spheroid set-ups are a more appropriate setting for testing of our models' applicability domain [77,78]. Moreover, investigating whether different OSR or DDR compounds could be described by these models would be an interesting expansion. Based on previous research where models were applied to different OSR-invoking compounds [39] and our experience in translating a model developed for cisplatin [27] to etoposide, we do not expect major changes in the model structure (in other words, in molecular interactions) but at least some changes in the parameter values are likely required (i.e., the magnitude of the interactions).

We do not expect major problems when expanding these models to different concentrations in the adaptive region, as we already applied them to 6 or 8 different doses of etoposide or DEM. Using the curves for the estimated initial stress levels (S3 Fig.), one could extrapolate the required stress value for the desired dose. However, for much higher concentrations we do not expect our models to be valid anymore. We specifically developed our models to describe the adaptive response to stress, when cells can still recover. When many cells die the average stress pathway activation in that population is expected to go down because a large cellular fraction will be dying and become unresponsive at stress pathway level [79], which affects the response observed in the population. This problem might be circumvented by investigating responses on a single-cell level, which would require a modified experimental design to enable single-cell tracking over time. In such efforts, we recommend combining investigation of stress pathways with death pathways that become substantially activated for high compound doses. This would provide quantitative insight in this interplay and a comprehensive view on DILI.

Another unresolved question is how well our models would perform to describe the dynamics in different cell types. We developed the models based on data in HepG2 cells. HepG2 cells are a commonly used cell type to represent liver cells but it is a carcinoma cell line and therefore not fully representative for hepatocytes. Among other things, they proliferate continuously and have low levels of metabolising enzymes [80]. Hepatocyte-like cells (HLC) differentiated from human-induced pluripotent stem cells are considered more similar to human liver tissue than HepG2 cells, although not as similar as Primary Human Hepatocytes (PHH) [81]. Unfortunately, time-dynamic protein data is not yet available for these cell types, although recently developed HLC reporter cell lines allow for such dynamic data [82]. Such data can be used in the future to apply our models to. We expect our models to be applicable to different liver cell types since etoposide activates the DNA damage response also in PHH [83] and DEM induces the oxidative stress response also in PHH [83,84] and HLC [82].

As mentioned before, cellular responses to stress are much more complex than considered within the models we have developed during this study. Despite several limitations, our models can simulate most of the data on the activation of both OSR and DDR-related proteins caused by exposure to DEM or etoposide. Moreover, the modelling implicated plausible key players in the regulation of crosstalk between the OSR and DDR pathways. Furthermore, *in silico* knockdown predictions largely agreed with previously published experimental data, indicating model validity. Future research integrating experimental data into computational models will be required to further elucidate the (crosstalk between) OSR and DDR pathways. Nevertheless, our current models could already be connected to other models that predict cell fate or adversity based on expression of specific proteins. In this way, our models can play a valuable role in the prediction of adverse reactions such as DILI.

Methods

In order to develop our p53-NRF2 pathway models, we used previously published data of Wijaya et al. [49]. In brief, Wijaya et al. [49] performed a screen with HepG2 BAC-GFP reporter cell lines [85] exposed to eight concentrations of various compounds. They used live fluorescence-based imaging of reporter activity for 60 hours with a time interval of 90 (BTG2, MDM2, p21 and p53) or 120 (NRF2 and SRXN1) minutes to monitor abundance of relevant proteins in the DDR and OSR pathways. Here, we use the data generated upon exposure to topoisomerase II inhibitor etoposide and oxidative stressor diethyl maleate (DEM) with the BTG2, MDM2, NRF2, p21, p53 and SRXN1 cell lines. For a detailed description of the experimental procedures, we refer to the Methods section in [49].

Data analysis

Viability data To select suitable concentrations for our modelling, we used the cell viability data generated by Wijaya et al. [49]. Specifically, we analysed the cell count, Annexin V (AnV) and Propidium Iodide (PI) measurements of all 11 reporters used in the screen (i.e. ATF4, BIP, CHOP, XBP1, NRF2, SRXN1, HMOX1, p53, MDM2, BTG2 and p21) that were exposed to DEM and etoposide. For each reporter, at least 3 replicates were available (for p21 4 replicates), resulting in a total of 34 time-course measurements that were used to compute the cell count, PI and AnV data for each treatment. To account for differences in the number of cells at the start of the experiment, we divided the number of cells by the number of cells at the first time point, i.e., each well started with a relative cell count of 1. To evaluate the amount of cell death, we used the fraction AnV- or PI-positive cells. A cell is regarded as AnV- or PI-positive when the AnV or PI area is at least 10 per cent of the total 2-pixel dilated nucleus area. Based on a decreasing cell count and a fraction of AnV-positive cells exceeding 0.15 for the two highest concentrations of etoposide, we omitted this subset of the data for model calibration.

GFP data Using the green fluorescent protein (GFP) intensities obtained after image analysis by Wijaya et al. [49] as representative for the abundance of labelled proteins, we calculated the geometric mean of the GFP measurements of all individual cells per image. Similarly, we also obtained the GFP intensity for wells treated with 0.01 μ M DMSO in this way. This control value corresponding to the same plate and time point was subtracted from the GFP intensity of each treatment to correct for background signal.

Next, we applied further normalisation to the background-corrected data using the following formula:

x

$$n_{ormalised} = \frac{x - x_{min}}{x_{mean} - x_{min}},$$
(1)

with *x* a measured value, x_{min} the minimally measured value among all treatments (see Table 1 in [49] for an overview) and time points of that plate, and x_{mean} the average value for the GFP intensity of all treatments and time points for that plate. With this approach, we obtained a relatively uniform scale between the replicates, avoiding potential major influence of one outlier that shifts all other values, which can happen for the commonly used min-max normalisation.

For every plate, the measurements were performed at slightly different time points due to the time required for experimental procedures. Interpolation of the data ensured suitability for our parameter fitting process (explained below). We interpolated the data to time points in the interval 0 to 62 hours with time steps of 1 hour with a natural cubic spline using the spline function of the stats package in R [86]. To obtain values for the initial time point(s), the data had to be extrapolated, because the initial measurements did not occur exactly at time point 0. We used the same natural spline approach for extrapolation but only used extrapolated values within 0.5 hours of the first measured time point to make sure the generated estimates would be close to the initial measurement. The mean of the extrapolated GFP intensities at the initial time point for each state variable was used as the initial state value for the associated model variable (see model explanation below) in the simulations. As for the AnV, PI, and cell count data, for p21 there were 4 replicates available and for all other reporters, the data contained 3 replicates.

Mathematical modelling

Base model DDR We used the DDR model of Heldring et al. [48] as base for the combined p53-NRF2 model designed in the current study. Compared to that model, we discarded the scaling and offset parameters, changed the initial states corresponding to the values in the etoposide data and recalibrated the parameters. The model contains seven state variables: S_{DD} (DNA damage inflicting stress), *DD* (DNA damage), p_{53} (unphosphorylated p53 protein), p_{53p} (phosphorylated p53 protein), M_2 (MDM2 protein), B_2 (BTG2 protein) and p_{21} (p21 protein)). The total p53, which is what the p53 reporter measures, is the sum of p_{53} and p_{53p} . The base model consists of the following system of equations:

$$\frac{dS_{DD}}{dt} = -\tau_D \cdot S_{DD}, \qquad (2)$$

$$\frac{dDD}{dt} = b_{DD} + S_{DD} - d_{DD, p_{53p}} \cdot p_{53p} \cdot DD,$$
(3)

$$\frac{dp_{53}}{dt} = b_{p_{53}} + r_{dp} \cdot p_{53p} - r_p \cdot p_{53} \cdot DD - d_{p_{53}} \cdot \left(1 + d_{p_{53},M_2} \cdot M_2\right) \cdot p_{53}, \tag{4}$$

$$\frac{dp_{53p}}{dt} = r_p \cdot p_{53} \cdot DD - r_{dp} \cdot p_{53p} - d_{p_{53p}} \cdot \left(1 + d_{p_{53p},M_2} \cdot M_2\right) \cdot p_{53p},$$
(5)

$$\frac{dM_2}{dt} = b_{M_2} + \frac{V_{M_2} \cdot p_{53p}^4}{Km_{M_2}^4 + p_{53p}^4} - d_{M_2} \cdot M_2, \qquad (6)$$

$$\frac{dB_2}{dt} = b_{B_2} + \frac{V_{B_2} \cdot p_{53p}^4}{Km_{B_2}^4 + p_{53p}^4} - d_{B_2} \cdot B_2, \qquad (7)$$

$$\frac{dp_{21}}{dt} = b_{p_{21}} + \frac{V_{p_{21}} \cdot p_{53p}^4}{Km_{p_{21}}^4 + p_{53p}^4} - d_{p_{21}} \cdot p_{21} \,. \tag{8}$$

In this system, τ_D is the decay rate for the DNA damage stress, b_X is the basal production rate for state variable *X*, d_X is the basal degradation rate of state variable *X*, d_{X,M_2} is the factor by which MDM2 promotes degradation of *X*, r_p is the phosphorylation rate of p53, r_{dp} is the

dephosphorylation rate of p53p, V_X is the maximal speed of X production (which depends on TF p53p), and Km_X is the amount of TF (i.e., p53p in this model) for which the production rate of X is half-maximal. Note that the Hill exponent for the non-linear production of p53 targets is set to 4, because phosphorylated p53 organizes in tetramers before binding to DNA.

Base model OSR We developed a simplified model of NRF2 signalling as a base for the subsequent combined p53-NRF2 model. The base NRF2 model contains the five state variables S_{OS} (oxidative stress), K_1 (KEAP1 protein), K_{1mod} (modified KEAP1 protein), N_2 (NRF2 protein) and S_1 (SRXN1 protein), and consists of the following equations:

$$\frac{dS_{OX}}{dt} = -\tau_O \cdot S_{OX}, \qquad (9)$$

$$\frac{dK_1}{dt} = b_{K_1} + \frac{V_{K_1} \cdot N_2^{n_{K_1}}}{Km_{K_1}^{n_{K_1}} + N_2^{n_{K_1}}} - r_m \cdot S_{OX} \cdot K_1 + r_{um} \cdot K_{1mod} - d_{K_1} \cdot K_1,$$
(10)

$$\frac{dK_{1mod}}{dt} = r_m \cdot S_{OX} \cdot K_1 - r_{um} \cdot K_{1mod} - d_{K_1} \cdot K_{1mod}, \qquad (11)$$

$$\frac{dN_2}{dt} = b_{N_2} - \frac{Vd_{N_2} \cdot K_1 \cdot N_2}{Km_{N_2} + N_2} - d_{N_2} \cdot N_2, \qquad (12)$$

$$\frac{dS_1}{dt} = b_{S_1} + \frac{V_{S_1} \cdot N_2^{n_{S_1}}}{Km_{S_1}^{n_{S_1}} + N_2^{n_{S_1}}} - d_{S_1} \cdot S_1 \,. \tag{13}$$

In these equations, as in the DDR model, b_X is the basal production for state variable Xand d_X is the basal degradation rate of state variable X, V_X is the maximal speed of X production (which depends on NRF2 here), and Km_X is the amount of TF (i.e., NRF2 in this model) for which the production or degradation rate of X is half-maximal. Furthermore, τ_O is the decay rate for the oxidative stress, r_m is the (stress-dependent) modification rate of KEAP1, r_{um} is the unmodification rate of modified KEAP1, Vd_{N_2} is the maximal KEAP1-dependent degradation rate of NRF2, n_X is the Hill coefficient for the non-linear production of X.

Crosstalk model etoposide To develop our crosstalk models, we combined the models of individual pathways described above and included several modes of crosstalk. With our first set of crosstalk models, aiming to describe the cellular response to etoposide, we focused on the impact of the DDR on the OSR (but not vice versa). Therefore, we included two modes of crosstalk suggested by literature. The first one describes p21-mediated inhibition of the KEAP1-dependent degradation of NRF2 [40], which we implemented by replacing Eq 12 with

$$\frac{dN_2}{dt} = b_{N_2} - \frac{Vd_{N_2} \cdot K_1 \cdot N_2}{(Km_{N_2} + C_{p_{21}} \cdot p_{21}) + N_2} - d_{N_2} \cdot N_2, \qquad (14)$$

with $C_{p_{21}}$ as a new parameter representing the strength of p21-mediated inhibition. The second mode of crosstalk is the inhibition of NRF2-dependent production of KEAP1 via p53p [41], which we implemented by replacing Eq 10 with

$$\frac{dK_1}{dt} = b_{K_1} + \frac{V_{K_1} \cdot N_2^{n_{K_1}}}{\left(Km_{K_1} + C_{p_{53p}} \cdot p_{53p}\right)^{n_{K_1}} + N_2^{n_{K_1}}} - r_m \cdot S_{OX} \cdot K_1 + r_{um} \cdot K_{1mod} - d_{K_1} \cdot K_1 , \quad (15)$$

with new parameter $C_{p_{53p}}$, representing the strength of p53p-mediated inhibition.

Moreover, we removed the variables for oxidative stress (S_{OS}) and modified KEAP1 (K_{1mod}) from this model, i.e., we do not consider etoposide to directly induce oxidative stress. We have developed three model variants to describe crosstalk upon exposure to etoposide, one for each mode of crosstalk and one for a combination of the two. An overview of the equation numbers forming each model variant and the quality of the simulation fits can be found in Table 1.

Crosstalk model DEM With our second set of crosstalk models, aiming to describe the cellular response to DEM, we focused on the impact of the OSR on the DDR (but not vice versa). We included several modes of crosstalk suggested by literature. First, to simulate DNA damage caused by oxidative stress (due to e.g. reactive oxygen species) [42], we replaced Eq 3 by:

$$\frac{dDD}{dt} = b_{DD} + S_{DD} + \frac{sc_S \cdot S_{OX}}{Km_S + S_{OX}} - d_{DD,p_{53p}} \cdot p_{53p} \cdot DD,$$
(16)

in which sc_S and Km_S are required as additional parameters, respectively representing the maximal rate of DNA damage induction by oxidative stress and the level of oxidative stress for which this rate is half-maximal. Moreover, we set S_{DD} to 0 for all simulations without 'direct' DNA damage stress.

To model a second mode of crosstalk, i.e., NRF2-dependent MDM2 production [43,44], we replaced Eq 6 by:

$$\frac{dM_2}{dt} = b_{M_2} + \frac{V_{M_2} \cdot p_{53p}^4}{Km_{M_2}^4 + p_{53p}^4} + \frac{V_{M_2,N_2} \cdot N_2^{n_{M_2}}}{Km_{M_2,N_2}^{n_{M_2}} + N_2^{n_{M_2}}} - d_{M_2} \cdot M_2 , \qquad (17)$$

where the new *Km* and *V* parameters (following the prior mentioned naming convention) describe the Hill function leading to additional non-linear MDM2 production.

To model a third mode of crosstalk, i.e., inhibition of basal p53 degradation via the NRF2 target NQO1 [45,54,55], we replaced Eq 4 with:

$$\frac{dp_{53}}{dt} = b_{p_{53}} + r_{dp} \cdot p_{53p} - r_p \cdot p_{53} \cdot DD - d_{p_{53}} \cdot \left(\frac{1}{1 + C_{S_1} \cdot S_1} + d_{p_{53},M_2} \cdot M_2\right) \cdot p_{53}.$$
 (18)

Here the parameter C_{S_1} represents the strength of inhibition. Note that we implemented the effect of NQO1 via SRXN1 rather than NQO1. This is because NQO1 is lacking from our OSR model due to absence of measurements of this NRF2 target. HepG2 transcriptomics data shows that NQO1 and SRXN1 are coregulated [57]. Therefore, we expect NQO1 to exhibit similar dynamics as SRXN1, and included it directly as such in our model.

To describe the DEM-induced DDR, we also considered adaptation of the MDM2dependent p53 degradation from a mass-action term to a non-linear Hill function (with new parameters Km_{p_{53},M_2} , Km_{p_{53p},M_2} , n_1 and n_2), thus replacing Eqs 4 and 5 with:

$$\frac{dp_{53}}{dt} = b_{p_{53}} + r_{dp} \cdot p_{53p} - r_p \cdot p_{53} \cdot DD - d_{p_{53}} \left(1 + \frac{d_{p_{53},M_2} \cdot M_2^{n_1}}{Km_{p_{53},M_2}^{n_1} + M_2^{n_1}} \right) \cdot p_{53},$$
(19)

and:

$$\frac{dp_{53p}}{dt} = r_p \cdot p_{53} \cdot DD - r_{dp} \cdot p_{53p} - d_{p_{53p}} \left(1 + \frac{d_{p_{53p},M_2} \cdot M_2^{n_2}}{Km_{p_{53p},M_2}^{n_2} + M_2^{n_2}} \right) \cdot p_{53p} \,. \tag{20}$$

Combining the adapted MDM2-dependent degradation of p53 and the inhibition of the basal degradation via NQO1 gives the following ODE for unphosphorylated p53:

$$\frac{dp_{53}}{dt} = b_{p_{53}} + r_{dp} \cdot p_{53p} - r_p \cdot p_{53} \cdot DD - d_{p_{53}} \left(\frac{1}{1 + C_{S_1} \cdot S_1} + \frac{d_{p_{53},M_2} \cdot M_2^{n_1}}{Km_{p_{53},M_2}^{n_1} + M_2^{n_1}} \right) \cdot p_{53}.$$
(21)

Based on the above 4 changes to the combined model, we implemented six different variants of the crosstalk model upon DEM exposure. In each of the models, we consider direct DNA damage to be absent, i.e. $S_{dd} = 0$. An overview of the equation numbers forming each model variants can be found in Table 2.

Two-directional crosstalk models Besides the above one-directional crosstalk models (either crosstalk from OSR to DDR, or vice versa), we also considered two crosstalk models taking into account both directions. First, in model M-CE, we added the crosstalk caused by the activation of the OSR to model M-E1, consisting of Eqs 2–5, 7, 8, 10, 13, 14, and 17. Second, in model M-CD, we added the crosstalk caused by DNA damage to model M-D4, consisting of Eqs 7–11, 13, 14, 16, 17, 19, and 20. As before, for model M-CE, S_{ox} and K_{1mod} was set to 0 and for model M-CD S_{dd} was set to 0. We did not recalibrate or change any of the parameters for these models. To make sure the models would still start at steady state, we first ran them without stress to converge to a steady state and after that applied the stress to the model.

Modelling of knockdowns To simulate knockdown of specific genes, we used models M-CE and M-CD as a basis, and reduced all production terms by 70% in each knockdown simulation. For example, for a knockdown of p53 in model M-CE, this results in:

$$\frac{dp_{53}}{dt} = 0.3 \cdot (b_{p_{53}}) + r_{dp} \cdot p_{53p} - r_p \cdot p_{53} \cdot DD - d_{p_{53}} (1 + d_{p_{53},M_2} \cdot M_2) \cdot p_{53}, \qquad (22)$$

instead of Eq 4. This way the production rate is reduced to 30% of its normal value without affecting the phosphorylation or degradation of p53. Similarly, knocking down p21 in the same model results in:

$$\frac{dp_{21}}{dt} = 0.3 \cdot \left(b_{p_{21}} + \frac{V_{p_{21}} \cdot p_{53p}^4}{Km_{p_{21}}^4 + p_{53p}^4} \right) - d_{p_{21}} \cdot p_{21} , \qquad (23)$$

instead of Eq 8. Here both the basal production and the MDM2-dependent production rates are reduced but the degradation of p21 is unaffected. We performed knockdown simulations in model M-CD in the same manner. Similar to the two-directional crosstalk models, we first let the models converge to a steady state with knockdown but without stress, and next applied stress to the models.

Parameter estimation The parameters of the models were fitted using an in-house fitting script, based on a maximum likelihood approach, previously described in [27,79,87]. In brief, it uses the least-squares method of the SciPy package in Python [88], in combination with sensitivity equations to find the path of the steepest descent towards the optimum [89]. With Latin Hypercube sampling different sets of starting values are generated [90]. The parameter set generating the lowest cost was deemed the optimum.

To ensure finding of the global optimum, a sufficiently large number of initial guesses for parameter sets should be used. For the base models, we used 200 sets, while for the crosstalk models we let the number of sets depend on the number of parameters to be estimated.

Specifically, we used 50 sets when calibrating models E1-3 and 100 when fitting models D1-6. Every time we checked if the lowest cost was reached by multiple sets (at least three) to gain confidence that this was indeed the optimum.

We predefined some of the initial values of variables directly based on the data rather than estimating them with parameter calibration because we had direct estimates available based on extrapolation of the data. Specifically, we only estimated the initial values for DNA damage, p53, phosphorylated p53 and unmodified KEAP1 during parameter estimation. For the other, directly measured variables (i.e., MDM2, p21, BTG2, NRF2 and SRXN1), we used the extrapolated value at the 0 hour timepoint as the initial state. The initial value for modified KEAP1 was set to 0.

As an additional constraint on the parameters, we defined steady state constraints in the absence of stress similar to Rosenblatt et al. [91], to ensure the model always starts in a steady state. We achieved this by selecting one parameter from each equation, to be defined by a steady state constraint. These constraints are determined by setting the ODE equal to 0, for a stress with value 0 and the initial state values for the other variables, and then solving this equation for the selected parameter. For equations starting in steady state by definition, i.e. for the stress (S_{DD} and S_{OS}) and modified KEAP1 (K_{1mod}), no additional constraints were applied to the parameters in those equations.

The lower bound for the fitted parameters was set to 0 and the upper bound to 1000, except for Hill coefficients, which we bounded between 1 and 10. The parameters have arbitrary units because GFP measurements only provide information on changes in relative abundance.

Model comparison Because we created various model variants, we wanted to formally compare how well they described the experimental data. When the number of parameters is the same, one can compare the models by comparing how much each model deviates from the measurements, for example by computing the Residual Squared Sum (RSS)

$$RSS = \sum_{i=1}^{N} (y_i - f(t_i))^2, \qquad (24)$$

with y_i the measurement at time point *i*, $f(t_i)$ the simulation value and *N* the number of time points.

However, when the number of parameters differs, a penalty needs to be included for the number of parameters in order to prevent overfitting. To this end, one can use the Akaike Information Criterion (AIC) [92] or the Bayesian Information Criterion (BIC) [93]. Considering normally distributed residuals, one can use the following expressions to compute the AIC and BIC:

$$AIC = n \cdot \ln\left(\frac{RSS}{n}\right) + 2K,$$
(25)

$$BIC = n \cdot \ln\left(\frac{RSS}{n}\right) + \ln(n)K,$$
(26)

with *n* the number of observations, *K* the number of parameters and the *RSS* as defined above. For sample sizes larger than 7, this leads to a larger penalty for an increasing number of parameters in BIC compared to AIC. Because AIC or BIC values do not mean anything on their own, but only the difference in AIC or BIC between two models is relevant, we selected the model with the lowest AIC or BIC as the best model to describe the data.

Sensitivity analysis Once we selected the best model for each drug, we performed a local sensitivity analysis on these models. Specifically, we increased and decreased each parameter separately by 20% and simulated the model for each adapted parameter. Here, we first ran the model without stress to converge to a steady state, and we next ran simulations with stress to the model. To obtain a measure for the effect of changing the parameter, for each model variable the absolute value of its difference between simulations with increased and decreased parameters was summed over all concentrations and time points (evaluated at 1-hour time points; S11A and S12A Figs).

Used software and data and code availability

For data analysis and data representation, we used R (version 4.3.3) [86] and the tidyverse package (version 2.0.0) [94]. For model simulations, we used the deSolve package (version 1.40) [95] in R. For parameter optimization we used Python (version 3.6.7). Schematic overviews of the pathways were made using Cytoscape [96]. All code to simulate the models and generate the figures is available at https://doi.org/10.5281/zenodo.14844666.

Supporting information

S1 Fig. Viability data for eight concentrations of etoposide. (PDF)

S2 Fig. DDR model describes data from HepG2 cells exposed to etoposide. (PDF)

S3 Fig. Initial value for the estimated stress levels for each compound concentration. (PDF)

S4 Fig. HepG2 cell viability data for eight concentrations of DEM. (PDF)

S5 Fig. OSR model describes data from HepG2 cells exposed to DEM when NRF2 stimulates KEAP1 production.

(PDF)

S6 Fig. Expression of p53 up to 20 hours after exposure to DEM. (PDF)

S7 Fig. Experimental data for BTG2 and p21 intensity after exposure of HepG2 cells to eight concentrations of DEM. (PDF)

S8 Fig. Crosstalk models to describe etoposide-induced OSR activity. (PDF)

S9 Fig. Crosstalk models to describe DEM-induced DDR activity. (PDF)

S10 Fig. Crosstalk model to describe DEM-induced DDR activity with altered MDM2 production.

(PDF)

S11 Fig. Sensitivity analysis for model M-E1. (PDF)

S12 Fig. Sensitivity analysis for model M-D4. (PDF) S13 Fig. Best DDR-OSR crosstalk models. (PDF) S14 Fig. In silico knockdown predictions on the basis of crosstalk models. (PDF) S15 Fig. In silico knockdown predictions for p21. (PDF) S1 Table. Parameter values for the DDR base model. (PDF) S2 Table. Parameter values for the OSR base models with and without NRF2-dependent **KEAP1** production. (PDF) S3 Table. New or changed parameters for models M-E1, M-E2 and M-E3. (PDF) S4 Table. New or changed parameters for models M-D1, M-D2 and M-D3. (PDF) S5 Table. New or changed parameters for models M-D4, M-D5 and M-D6. (PDF)

Acknowledgments

We would like to thank the members of the SysBioTop-Moving Consortium for scientific discussions related to this work.

Author contributions

Conceptualization: Elsje J. Burgers, Raju P. Sharma, Joost B. Beltman.

Data curation: Lukas S. Wijaya.

Formal analysis: Elsje J. Burgers.

Funding acquisition: Bob van de Water, Joost B. Beltman.

Investigation: Elsje J. Burgers.

Methodology: Elsje J. Burgers, Raju P. Sharma, Carl Joshua S. Eugenio, Muriel M. Heldring.

Supervision: Bob van de Water, Joost B. Beltman.

Visualization: Elsje J. Burgers.

Writing - original draft: Elsje J. Burgers.

Writing – review & editing: Elsje J. Burgers, Raju P. Sharma, Carl Joshua S. Eugenio, Muriel M. Heldring, Lukas S. Wijaya, Bob van de Water, Joost B. Beltman.

References

1. Watkins PB. Drug safety sciences and the bottleneck in drug development. Clin Pharmacol Ther. 2011;89(6):788–90. https://doi.org/10.1038/clpt.2011.63 PMID: 21593756

- Siramshetty VB, Nickel J, Omieczynski C, Gohlke B-O, Drwal MN, Preissner R. WITHDRAWN–a resource for withdrawn and discontinued drugs. Nucl Acids Res. 2016;44(D1):D1080-6. https://doi.org/10.1093/nar/gkv1192 PMID: 26553801
- Craveiro NS, Lopes BS, Tomás L, Almeida SF. Drug withdrawal due to safety: a review of the data supporting withdrawal decision. Curr Drug Saf. 2020;15(1):4–12. https://doi.org/10.2174/1574886314666191004092520 PMID: 31584381
- Van Norman GA. Limitations of animal studies for predicting toxicity in clinical trials: is it time to rethink our current approach? JACC Basic Transl Sci. 2019;4(7):845–54. https://doi.org/10.1016/j.jacbts.2019.10.008 PMID: 31998852
- Krewski D, Acosta D Jr, Andersen M, Anderson H, Bailar JC 3rd, Boekelheide K, et al. Toxicity testing in the 21st century: a vision and a strategy. J Toxicol Environ Health B Crit Rev. 2010;13(2–4):51–138. https://doi.org/10.1080/10937404.2010.483176 PMID: 20574894
- Kiani AK, Pheby D, Henehan G, Brown R, Sieving P, Sykora P, et al. Ethical considerations regarding animal experimentation. J Prev Med Hyg. 2022;63(2 Suppl 3):E255–66. https://doi.org/10.15167/2421-4248/jpmh2022.63.2S3.2768 PMID: 36479489
- Di Zeo-Sánchez DE, Segovia-Zafra A, Matilla-Cabello G, Pinazo-Bandera JM, Andrade RJ, Lucena MI, et al. Modeling drug-induced liver injury: current status and future prospects. Expert Opin Drug Metab Toxicol. 2022;18(9):555–73. https://doi.org/10.1080/17425255.2022.2122810 PMID: 36107152
- Pognan F, Beilmann M, Boonen HCM, Czich A, Dear G, Hewitt P, et al. The evolving role of investigative toxicology in the pharmaceutical industry. Nat Rev Drug Discov. 2023;22(4):317–35. https://doi.org/10.1038/s41573-022-00633-x PMID: 36781957
- Klipp E, Liebermeister W. Mathematical modeling of intracellular signaling pathways. BMC Neurosci. 2006;7(Suppl 1):S10. https://doi.org/10.1186/1471-2202-7-S1-S10 PMID: 17118154
- Daun S, Rubin J, Vodovotz Y, Clermont G. Equation-based models of dynamic biological systems. J Crit Care. 2008;23(4):585–94. https://doi.org/10.1016/j.jcrc.2008.02.003 PMID: 19056027
- 11. Kültz D. Molecular and evolutionary basis of the cellular stress response. Annu Rev Physiol. 2005;67(1):225–57. https://doi.org/10.1146/annurev.physiol.67.040403.103635
- lorga A, Dara L, Kaplowitz N. Drug-induced liver injury: cascade of events leading to cell death, apoptosis or necrosis. Int J Mol Sci. 2017;18(5):1018. https://doi.org/10.3390/ijms18051018 PMID: 28486401
- Kuijper IA, Yang H, Van De Water B, Beltman JB. Unraveling cellular pathways contributing to drug-induced liver injury by dynamical modeling. Expert Opin Drug Metab Toxicol. 2017;13(1):5–17. https://doi.org/10.1080/17425255.2017.1234607 PMID: 27609146
- Malik-Sheriff RS, Glont M, Nguyen TVN, Tiwari K, Roberts MG, Xavier A, et al. BioModels—15 years of sharing computational models in life science. Nucl Acids Res. 2019;48(D1):D407–D415. https://doi.org/10.1093/nar/gkz1055
- Kruiswijk F, Labuschagne CF, Vousden KH. p53 in survival, death and metabolic health: a lifeguard with a licence to kill. Nat Rev Mol Cell Biol. 2015;16(7):393–405. https://doi.org/10.1038/nrm4007 PMID: 26122615
- Hafner A, Bulyk ML, Jambhekar A, Lahav G. The multiple mechanisms that regulate p53 activity and cell fate. Nat Rev Mol Cell Biol. 2019;20(4):199–210. https://doi.org/10.1038/s41580-019-0110-x PMID: 30824861
- Yagishita Y, Chartoumpekis DV, Kensler TW, Wakabayashi N. NRF2 and the Moirai: life and death decisions on cell fates. Antioxid Redox Signal. 2023;38(7–9):684–708. https://doi.org/10.1089/ars.2022.0200 PMID: 36509429
- 18. Ghanim BY, Qinna NA. Nrf2/ARE axis signalling in hepatocyte cellular death. Mol Biol Rep. 2022;49(5):4039–53. https://doi.org/10.1007/s11033-022-07125-6 PMID: 35020121
- 19. Carusillo A, Mussolino C. DNA damage: from threat to treatment. Cells. 2020;9(7):1665. https://doi.org/10.3390/cells9071665 PMID: 32664329
- 20. Zhou B, Elledge S. Checkpoints in perspective. Nature. 2000;408(11):433-9.
- Blackford AN, Jackson SP. ATM, ATR, and DNA-PK: the trinity at the heart of the DNA damage response. Mol Cell. 2017;66(6):801–17. https://doi.org/10.1016/j.molcel.2017.05.015 PMID: 28622525
- 22. Riley T, Sontag E, Chen P, Levine A. Transcriptional control of human p53-regulated genes. Nat Rev Mol Cell Biol. 2008;9(5):402–12. https://doi.org/10.1038/nrm2395 PMID: 18431400
- **23.** Michael D, Oren M. The p53-Mdm2 module and the ubiquitin system. Semin Cancer Biol. 2003;13(1):49–58. https://doi.org/10.1016/s1044-579x(02)00099-8 PMID: 12507556

- Lahav G, Rosenfeld N, Sigal A, Geva-Zatorsky N, Levine AJ, Elowitz MB, et al. Dynamics of the p53-Mdm2 feedback loop in individual cells. Nat Genet. 2004;36(2):147–50. https://doi.org/10.1038/ng1293 PMID: 14730303
- 25. Kim E, Kim J-Y, Lee J-Y. Mathematical modeling of p53 pathways. Int J Mol Sci. 2019;20(20):5179. https://doi.org/10.3390/ijms20205179 PMID: 31635420
- Eliaš J, Macnamara CK. Mathematical modelling of p53 signalling during DNA damage response: a survey. Int J Mol Sci. 2021;22(19):10590. https://doi.org/10.3390/ijms221910590 PMID: 34638930
- Heldring MM, Wijaya LS, Niemeijer M, Yang H, Lakhal T, Le Dévédec SE, et al. Model-based translation of DNA damage signaling dynamics across cell types. PLoS Comput Biol. 2022;18(7):e1010264. https://doi.org/10.1371/journal.pcbi.1010264 PMID: 35802572
- Espinosa-Diez C, Miguel V, Mennerich D, Kietzmann T, Sánchez-Pérez P, Cadenas S, et al. Antioxidant responses and cellular adjustments to oxidative stress. Redox Biol. 2015;6:183–97. https://doi.org/10.1016/j.redox.2015.07.008 PMID: 26233704
- Itoh K, Wakabayashi N, Katoh Y, Ishii T, Igarashi K, Engel JD, et al. Keap1 represses nuclear activation of antioxidant responsive elements by Nrf2 through binding to the amino-terminal Neh2 domain. Genes Dev. 1999;13(1):76–86. https://doi.org/10.1101/gad.13.1.76 PMID: 9887101
- McMahon M, Itoh K, Yamamoto M, Hayes JD. Keap1-dependent proteasomal degradation of transcription factor Nrf2 contributes to the negative regulation of antioxidant response element-driven gene expression. J Biol Chem. 2003;278(24):21592–600. https://doi.org/10.1074/jbc.M300931200 PMID: 12682069
- Yamamoto M, Kensler TW, Motohashi H. The KEAP1-NRF2 system: a thiol-based sensor-effector apparatus for maintaining redox homeostasis. Physiol Rev. 2018;98(3):1169–203. https://doi.org/10.1152/physrev.00023.2017 PMID: 29717933
- 32. Tong KI, Kobayashi A, Katsuoka F, Yamamoto M. Two-site substrate recognition model for the Keap1-Nrf2 system: a hinge and latch mechanism. Biol Chem. 2006;387(10–11):1311–20. https://doi.org/10.1515/BC.2006.164 PMID: 17081101
- Tong KI, Padmanabhan B, Kobayashi A, Shang C, Hirotsu Y, Yokoyama S, et al. Different electrostatic potentials define ETGE and DLG motifs as hinge and latch in oxidative stress response. Mol Cell Biol. 2007;27(21):7511–21. https://doi.org/10.1128/MCB.00753-07 PMID: 17785452
- Horie Y, Suzuki T, Inoue J, Iso T, Wells G, Moore TW, et al. Molecular basis for the disruption of Keap1-Nrf2 interaction via Hinge & Latch mechanism. Commun Biol. 2021;4(1):576. https://doi.org/10.1038/s42003-021-02100-6 PMID: 33990683
- 35. Itoh K, Chiba T, Takahashi S, Ishii T, Igarashi K, Katoh Y, et al. An Nrf2/small Maf heterodimer mediates the induction of phase II detoxifying enzyme genes through antioxidant response elements. Biochem Biophys Res Commun. 1997;236(2):313–22. https://doi.org/10.1006/bbrc.1997.6943 PMID: 9240432
- **36.** Raghunath A, Sundarraj K, Nagarajan R, Arfuso F, Bian J, Kumar AP, et al. Antioxidant response elements: discovery, classes, regulation and potential applications. Redox Biol. 2018;17:297–314. https://doi.org/10.1016/j.redox.2018.05.002 PMID: 29775961
- Zhang Q, Andersen ME. Dose response relationship in anti-stress gene regulatory networks. PLoS Comput Biol. 2007;3(3):e24. https://doi.org/10.1371/journal.pcbi.0030024 PMID: 17335342
- Khalil HS, Goltsov A, Langdon SP, Harrison DJ, Bown J, Deeni Y. Quantitative analysis of NRF2 pathway reveals key elements of the regulatory circuits underlying antioxidant response and proliferation of ovarian cancer cells. J Biotechnol. 2015;202:12–30. https://doi.org/10.1016/j.jbiotec.2014.09.027 PMID: 25449014
- Hiemstra S, Fehling-Kaschek M, Kuijper IA, Bischoff LJM, Wijaya LS, Rosenblatt M, et al. Dynamic modeling of Nrf2 pathway activation in liver cells after toxicant exposure. Sci Rep. 2022;12(1):7336. https://doi.org/10.1038/s41598-022-10857-x PMID: 35513409
- 40. Chen W, Sun Z, Wang X-J, Jiang T, Huang Z, Fang D, et al. Direct interaction between Nrf2 and p21(Cip1/WAF1) upregulates the Nrf2-mediated antioxidant response. Mol Cell. 2009;34(6):663–73. https://doi.org/10.1016/j.molcel.2009.04.029 PMID: 19560419
- Faraonio R, Vergara P, Di Marzo D, Pierantoni MG, Napolitano M, Russo T, et al. p53 suppresses the Nrf2-dependent transcription of antioxidant response genes. J Biol Chem. 2006;281(52):39776–84. https://doi.org/10.1074/jbc.M605707200 PMID: 17077087
- Singh A, Kukreti R, Saso L, Kukreti S. Oxidative stress: role and response of short guanine tracts at genomic locations. Int J Mol Sci. 2019;20(17):4258. https://doi.org/10.3390/ijms20174258 PMID: 31480304
- **43.** You A, Nam C-W, Wakabayashi N, Yamamoto M, Kensler TW, Kwak M-K. Transcription factor Nrf2 maintains the basal expression of Mdm2: an implication of the regulation of p53 signaling by Nrf2.

Arch Biochem Biophys. 2011;507(2):356–64. https://doi.org/10.1016/j.abb.2010.12.034 PMID: 21211512

- Todoric J, Antonucci L, Di Caro G, Li N, Wu X, Lytle NK, et al. Stress-activated NRF2-MDM2 cascade controls neoplastic progression in pancreas. Cancer Cell. 2017;32(6):824-839.e8. https://doi.org/10.1016/j.ccell.2017.10.011 PMID: 29153842
- **45.** Asher G, Lotem J, Cohen B, Sachs L, Shaul Y. Regulation of p53 stability and p53-dependent apoptosis by NADH quinone oxidoreductase 1. Proc Natl Acad Sci U S A. 2001;98(3):1188–93. https://doi.org/10.1073/pnas.98.3.1188 PMID: 11158615
- Konrath F, Mittermeier A, Cristiano E, Wolf J, Loewer A. A systematic approach to decipher crosstalk in the p53 signaling pathway using single cell dynamics. PLoS Comput Biol. 2020;16(6):e1007901. https://doi.org/10.1371/journal.pcbi.1007901 PMID: 32589666
- Pereira EJ, Burns JS, Lee CY, Marohl T, Calderon D, Wang L, et al. Sporadic activation of an oxidative stress-dependent NRF2-p53 signaling network in breast epithelial spheroids and premalignancies. Sci Signal. 2020;13(627):eaba4200. https://doi.org/10.1126/scisignal.aba4200 PMID: 32291314
- Heldring MM, Shaw AH, Beltman JB. Unraveling the effect of intra- and intercellular processes on acetaminophen-induced liver injury. NPJ Syst Biol Appl. 2022;8(1):27. https://doi.org/10.1038/s41540-022-00238-5 PMID: 35933513
- 49. Wijaya LS, Trairatphisan P, Gabor A, Niemeijer M, Keet J, Alcalà Morera A, et al. Integration of temporal single cell cellular stress response activity with logic-ODE modeling reveals activation of ATF4-CHOP axis as a critical predictor of drug-induced liver injury. Biochem Pharmacol. 2021;190:114591. https://doi.org/10.1016/j.bcp.2021.114591
- Hande KR. Etoposide: four decades of development of a topoisomerase II inhibitor. Eur J Cancer. 1998;34(10):1514–21. https://doi.org/10.1016/s0959-8049(98)00228-7 PMID: 9893622
- Maellaro E, Casini AF, Del Bello B, Comporti M. Lipid peroxidation and antioxidant systems in the liver injury produced by glutathione depleting agents. Biochem Pharmacol. 1990;39(10):1513–21. https://doi.org/10.1016/0006-2952(90)90515-m PMID: 2337408
- Casey W, Anderson S, Fox T, Dold K, Colton H, Morgan K. Transcriptional and physiological responses of HepG2 cells exposed to diethyl maleate: time course analysis. Physiol Genomics. 2002;8(2):115–22. https://doi.org/10.1152/physiolgenomics.00064.2001 PMID: 11875189
- Lee O-H, Jain AK, Papusha V, Jaiswal AK. An auto-regulatory loop between stress sensors INrf2 and Nrf2 controls their cellular abundance. J Biol Chem. 2007;282(50):36412–20. https://doi.org/10.1074/jbc.M706517200 PMID: 17925401
- 54. Asher G, Lotem J, Kama R, Sachs L, Shaul Y. NQO1 stabilizes p53 through a distinct pathway. Proc Natl Acad Sci U S A. 2002;99(5):3099–104. https://doi.org/10.1073/pnas.052706799 PMID: 11867746
- 55. Anwar A, Dehn D, Siegel D, Kepa JK, Tang LJ, Pietenpol JA, et al. Interaction of human NAD(P)H:quinone oxidoreductase 1 (NQO1) with the tumor suppressor protein p53 in cells and cell-free systems. J Biol Chem. 2003;278(12):10368–73. https://doi.org/10.1074/jbc.M211981200 PMID: 12529318
- 56. Asher G, Lotem J, Sachs L, Kahana C, Shaul Y. Mdm-2 and ubiquitin-independent p53 proteasomal degradation regulated by NQO1. Proc Natl Acad Sci U S A. 2002;99(20):13125–30. https://doi.org/10.1073/pnas.202480499 PMID: 12232053
- 57. Wijaya LS, Gabor A, Pot IE, van de Have L, Saez-Rodriguez J, Stevens JL, et al. A network-based transcriptomic landscape of HepG2 cells uncovering causal gene-cytotoxicity interactions underlying drug-induced liver injury. Toxicol Sci. 2024;198(1):14–30. https://doi.org/10.1093/toxsci/kfad121 PMID: 38015832
- Hiemstra S, Niemeijer M, Koedoot E, Wink S, Klip JE, Vlasveld M, et al. Comprehensive landscape of Nrf2 and p53 pathway activation dynamics by oxidative stress and DNA damage. Chem Res Toxicol. 2017;30(4):923–33. https://doi.org/10.1021/acs.chemrestox.6b00322 PMID: 27982581
- Chen W, Jiang T, Wang H, Tao S, Lau A, Fang D, et al. Does Nrf2 contribute to p53-mediated control of cell survival and death?. Antioxid Redox Signal. 2012;17(12):1670–5. https://doi.org/10.1089/ars.2012.4674 PMID: 22559194
- Karve TM, Rosen EM. B-cell translocation gene 2 (BTG2) stimulates cellular antioxidant defenses through the antioxidant transcription factor NFE2L2 in human mammary epithelial cells. J Biol Chem. 2012;287(37):31503–14. https://doi.org/10.1074/jbc.M112.367433 PMID: 22493435
- Wang S, Chan SY, Deng Y, Khoo BL, Chua SL. Oxidative stress induced by Etoposide anti-cancer chemotherapy drives the emergence of tumor-associated bacteria resistance to fluoroquinolones. J Adv Res. 2024;55:33–44. https://doi.org/10.1016/j.jare.2023.02.011 PMID: 36822389

- Rowe LA, Degtyareva N, Doetsch PW. DNA damage-induced reactive oxygen species (ROS) stress response in Saccharomyces cerevisiae. Free Radic Biol Med. 2008;45(8):1167–77. https://doi.org/10.1016/j.freeradbiomed.2008.07.018 PMID: 18708137
- Kang MA, So E-Y, Simons AL, Spitz DR, Ouchi T. DNA damage induces reactive oxygen species generation through the H2AX-Nox1/Rac1 pathway. Cell Death Dis. 2012;3(1):e249. https://doi.org/10.1038/cddis.2011.134 PMID: 22237206
- 64. Lingappan K. NF-κB in oxidative stress. Curr Opin Toxicol. 2018;7:81–6. https://doi.org/10.1016/j.cotox.2017.11.002
- 65. Gao W, Guo L, Yang Y, Wang Y, Xia S, Gong H, et al. Dissecting the crosstalk between Nrf2 and NF-κB response pathways in drug-induced toxicity. Front Cell Dev Biol. 2022;9:809952. https://doi.org/10.3389/fcell.2021.809952 PMID: 35186957
- 66. Janssens S, Tschopp J. Signals from within: the DNA-damage-induced NF-kappaB response. Cell Death Differ. 2006;13(5):773–84. https://doi.org/10.1038/sj.cdd.4401843 PMID: 16410802
- Picco V, Pagès G. Linking JNK activity to the DNA damage response. Genes Cancer. 2013;4(9–10):360–8. https://doi.org/10.1177/1947601913486347 PMID: 24349633
- Desaint S, Luriau S, Aude J-C, Rousselet G, Toledano MB. Mammalian antioxidant defenses are not inducible by H2O2. J Biol Chem. 2004;279(30):31157–63. https://doi.org/10.1074/jbc.M401888200 PMID: 15155764
- 69. Seki E, Brenner DA, Karin M. A liver full of JNK: signaling in regulation of cell function and disease pathogenesis, and clinical approaches. Gastroenterology. 2012;143(2):307–20. https://doi.org/10.1053/j.gastro.2012.06.004 PMID: 22705006
- 70. Hong Y, Boiti A, Vallone D, Foulkes NS. Reactive oxygen species signaling and oxidative stress: transcriptional regulation and evolution. Antioxidants (Basel). 2024;13(3):312. https://doi.org/10.3390/antiox13030312 PMID: 38539845
- 71. Hiemstra S, Ramaiahgari SC, Wink S, Callegaro G, Coonen M, Meerman J, et al. High-throughput confocal imaging of differentiated 3D liver-like spheroid cellular stress response reporters for identification of drug-induced liver injury liability. Arch Toxicol. 2019;93(10):2895–911. https://doi.org/10.1007/s00204-019-02552-0 PMID: 31552476
- **72.** Wang X, Liu Y, Han A, Tang C, Xu R, Feng L, et al. The NQO1/p53/SREBP1 axis promotes hepatocellular carcinoma progression and metastasis by regulating Snail stability. Oncogene. 2022;41(47):5107–20. https://doi.org/10.1038/s41388-022-02477-6 PMID: 36253445
- 73. Magnussen HM, Ahmed SF, Sibbet GJ, Hristova VA, Nomura K, Hock AK, et al. Structural basis for DNA damage-induced phosphoregulation of MDM2 RING domain. Nat Commun. 2020;11(1):2094. https://doi.org/10.1038/s41467-020-15783-y PMID: 32350255
- 74. Shvarts A, Steegenga WT, Riteco N, van Laar T, Dekker P, Bazuine M, et al. MDMX: a novel p53-binding protein with some functional properties of MDM2.. The EMBO Journal. 1996;15(19):5349–57. https://doi.org/10.1002/j.1460-2075.1996.tb00919.x
- 75. Gu J, Kawai H, Nie L, Kitao H, Wiederschain D, Jochemsen AG, et al. Mutual dependence of MDM2 and MDMX in their functional inactivation of p53. J Biol Chem. 2002;277(22):19251–4. https://doi.org/10.1074/jbc.C200150200 PMID: 11953423
- 76. Klein AM, de Queiroz RM, Venkatesh D, Prives C. The roles and regulation of MDM2 and MDMX: it is not just about p53. Genes Dev. 2021;35(9–10):575–601. https://doi.org/10.1101/gad.347872.120 PMID: 33888565
- 77. Cox CR, Lynch S, Goldring C, Sharma P. Current perspective: 3D spheroid models utilizing human-based cells for investigating metabolism-dependent drug-induced liver injury. Front Med Technol. 2020;2:611913. https://doi.org/10.3389/fmedt.2020.611913 PMID: 35047893
- 78. Kammerer S. Three-dimensional liver culture systems to maintain primary hepatic properties for toxicological analysis in vitro. Int J Mol Sci. 2021;22(19):10214. https://doi.org/10.3390/ijms221910214 PMID: 34638555
- 79. Yang H, Niemeijer M, van de Water B, Beltman JB. ATF6 Is a critical determinant of CHOP dynamics during the unfolded protein response. iScience. 2020;23(2):100860. https://doi.org/10.1016/j.isci.2020.100860 PMID: 32058971
- **80.** Guo L, Dial S, Shi L, Branham W, Liu J, Fang J-L, et al. Similarities and differences in the expression of drug-metabolizing enzymes between human hepatic cell lines and primary human hepatocytes. Drug Metab Dispos. 2011;39(3):528–38. https://doi.org/10.1124/dmd.110.035873 PMID: 21149542
- Gupta R, Schrooders Y, Hauser D, van Herwijnen M, Albrecht W, Ter Braak B, et al. Comparing in vitro human liver models to in vivo human liver using RNA-Seq. Arch Toxicol. 2021;95(2):573–89. https://doi.org/10.1007/s00204-020-02937-6 PMID: 33106934Gupta R, Schrooders Y, Hauser D, van Herwijnen M, Albrecht W, Ter Braak B, et al. Comparing in vitro human liver models to in vivo human liver using RNA-Seq. Arch Toxicol. 2021;95(2):573–89. https://doi.org/10.1007/s00204-020-02937-6 PMID: 33106934

- Snijders KE, Feher A, Tancos Z, Bock I, Teglasi A, van den Berk L, et al. Fluorescent tagging of endogenous Heme oxygenase-1 in human induced pluripotent stem cells for high content imaging of oxidative stress in various differentiated lineages. Archiv Toxicol. 2021;95(10):3285–302. https://doi.org/10.1007/s00204-021-03127-8.
- Callegaro G, Kunnen SJ, Trairatphisan P, Grosdidier S, Niemeijer M, den Hollander W, et al. The human hepatocyte TXG-MAPr: gene co-expression network modules to support mechanism-based risk assessment. Arch Toxicol. 2021;95(12):3745–75. https://doi.org/10.1007/s00204-021-03141-w PMID: 34626214
- Niemeijer M, Wiecek W, Fu S, Huppelschoten S, Bouwman P, Baze A, et al. Mapping interindividual variability of toxicodynamics using high-throughput transcriptomics and primary human hepatocytes from fifty donors. Environ Health Perspect. 2024;132(3):37005. https://doi.org/10.1289/EHP11891 PMID: 38498338
- 85. Wink S, Hiemstra S, Huppelschoten S, Danen E, Niemeijer M, Hendriks G, et al. Quantitative high content imaging of cellular adaptive stress response pathways in toxicity for chemical safety assessment. Chem Res Toxicol. 2014;27(3):338–55. https://doi.org/10.1021/tx4004038 PMID: 24450961
- 86. R Core Team. R: A language and environment for statistical computing; 2024. Available from: https://www.r-project.org/
- Yang H, Stel Wvd, Lee R, Bauch C, Bevan S, Walker P, et al. Dynamic modeling of mitochondrial membrane potential upon exposure to mitochondrial inhibitors. Front Pharmacol. 2021;12:679407. https://doi.org/10.3389/fphar.2021.679407.
- Virtanen P, Gommers R, Oliphant TE, Haberland M, Reddy T, Cournapeau D, et al. SciPy 1.0: fundamental algorithms for scientific computing in Python. Nat Methods. 2020;17(3):261–72. https://doi.org/10.1038/s41592-019-0686-2 PMID: 32015543
- Raue A, Schilling M, Bachmann J, Matteson A, Schelke M, Kaschek D, et al. Lessons learned from quantitative dynamical modeling in systems biology. PLoS ONE. 2013;8(9):e74335. https://doi.org/10.1371/journal.pone.0074335
- Owen AB. A central limit theorem for latin hypercube sampling. J Roy Statist Soc Ser B: Statist Methodol. 1992;54(2):541–51. https://doi.org/10.1111/j.2517-6161.1992.tb01895.x
- Rosenblatt M, Timmer J, Kaschek D. Customized steady-state constraints for parameter estimation in non-linear ordinary differential equation models. Front Cell Dev Biol. 2016;441. https://doi.org/10.3389/fcell.2016.00041 PMID: 27243005
- **92.** Akaike H. A new look at the statistical model identification. IEEE Trans Automat Contr. 1974;19(6):716–23. https://doi.org/10.1109/tac.1974.1100705
- **93.** Schwarz G. Estimating the dimension of a model. Ann Statist. 1978;6(2):. https://doi.org/10.1214/aos/1176344136
- 94. Wickham H, Averick M, Bryan J, Chang W, McGowan L, François R, et al. Welcome to the Tidyverse. JOSS. 2019;4(43):1686. https://doi.org/10.21105/joss.01686
- Soetaert K, Petzoldt T, Setzer RW. Solving differential equations inR: PackagedeSolve. J Stat Soft. 2010;33(9):1–25. https://doi.org/10.18637/jss.v033.i09
- 96. Shannon P, Markiel A, Ozier O, Baliga NS, Wang JT, Ramage D, et al. Cytoscape: a software environment for integrated models of biomolecular interaction networks. Genome Res. 2003;13(11):2498–504. https://doi.org/10.1101/gr.1239303 PMID: 14597658